Author's response to reviews

Title: The working alliance in a randomized controlled trial comparing online with face-to-face cognitive-behavioral therapy for depression

Authors:

Barbara Preschl (b.preschl@psychologie.uzh.ch)
Andreas Maercker (maercker@psychologie.uzh.ch)
Birgit Wagner (birgit.wagner@medizin.uni-leipzig.de)

Version: 4 Date: 30 November 2011

Author's response to reviews:

Dear Dr. Olino,

Many thanks for sending us the information that our manuscript is accepted for publication in principal. We have further received the information of the required format changes.

Following these instructions, we have removed the extra page from figure 1 and checked the manuscript for any typographical errors.

Since it was not totally clear to us in which way our tables should be changed we would as offered ask the editorial production team to make these changes.

We would like to thank you and your team for making the publication of our manuscript possible and further for all comments and advice through the publication process that was in our opinion amazingly rapid and efficient. Thank you very much!

Please let us know if you have any further questions.

Kind regards,

Barbara Preschl